Author’s response to reviews

Title: Effects of ‘plate model’ as a part of dietary intervention on rehabilitation of post myocardial infarction patients: study protocol for a randomized controlled trial

Authors:

Ranil Jayawardena (rani7@gmail.com)
Pasindu Fernando (pasindu9091@gmail.com)
Niroshan Lokunarangoda (niroshan_loku@yahoo.com)
Anidu Pathirana (anidupathirana@hotmail.com)

Version: 1 Date: 12 May 2017

Author’s response to reviews:

24/4/2017

Editors-in Chief

Doug Altman, University of Oxford, UK
Jeremy Grimshaw, Ottawa Health Research Institute, Canada

Trials

Dear Sirs,

Submission of revised manuscript

RE: TRLS-D-16-00855 - Effects of the ‘plate model’ as part of dietary intervention on modification of selected cardio-metabolic risk factors in post myocardial infarction patients: study protocol for a randomized controlled trial.

Thank you for your letter and the opportunity for resubmission. The constructive feedback from the journal was invaluable and we are now submitting the revised manuscript.
We have attached our responses to your comments made on a point by point basis and have indicated explicitly the ways in which we have changed the manuscript in line with the reviewers' comments. We have made necessary changes in the manuscript based on the suggestions.

Thank you for considering the paper for publication.

Yours Sincerely,

Dr. Ranil Jayawardena

Senior Lecturer and Consultant Clinical Nutritionist,

Department of Physiology,

Faculty of Medicine,

University of Colombo,